

UvA-DARE (Digital Academic Repository)

Adaptive foraging of pollinators fosters gradual tipping under resource competition and rapid environmental change

Terpstra, S.; Darcie Marquitti, F.M.; Vasconcelos, V.V.

DOI

10.1371/journal.pcbi.1011762

Publication date 2024

Document Version Final published version

Published in PLoS Computational Biology

License CC BY

Link to publication

Citation for published version (APA):

Terpstra, S., Darcie Marquitti, F. M., & Vasconcelos, V. V. (2024). Adaptive foraging of pollinators fosters gradual tipping under resource competition and rapid environmental change. *PLoS Computational Biology, 20*, Article e1011762. https://doi.org/10.1371/journal.pcbi.1011762

General rights

It is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), other than for strictly personal, individual use, unless the work is under an open content license (like Creative Commons).

Disclaimer/Complaints regulations

If you believe that digital publication of certain material infringes any of your rights or (privacy) interests, please let the Library know, stating your reasons. In case of a legitimate complaint, the Library will make the material inaccessible and/or remove it from the website. Please Ask the Library: https://uba.uva.nl/en/contact, or a letter to: Library of the University of Amsterdam, Secretariat, Singel 425, 1012 WP Amsterdam, The Netherlands. You will be contacted as soon as possible.

UvA-DARE is a service provided by the library of the University of Amsterdam (https://dare.uva.nl)



G OPEN ACCESS

Citation: Terpstra S, Marquitti FMD, Vasconcelos VV (2024) Adaptive foraging of pollinators fosters gradual tipping under resource competition and rapid environmental change. PLoS Comput Biol 20(1): e1011762. https://doi.org/10.1371/journal. pcbi.1011762

Editor: Mercedes Pascual, University of Chicago, UNITED STATES

Received: March 1, 2023

Accepted: December 15, 2023

Published: January 9, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1011762

Copyright: © 2024 Terpstra et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data and model are published online as: Terpstra, Sjoerd, Marquitti, Flávia M.D., & Vasconcelos, Vítor V. (2023). Model Implementation of: Adaptive Foraging of Pollinators **RESEARCH ARTICLE**

Adaptive foraging of pollinators fosters gradual tipping under resource competition and rapid environmental change

Sjoerd Terpstra^{1,2,3}, Flávia M. D. Marquitti^{4,5}, Vítor V. Vasconcelos^{3,6}*

1 Graduate School of Informatics, University of Amsterdam, Amsterdam, The Netherlands, 2 Institute for Marine and Atmospheric Research Utrecht, Utrecht University, Utrecht, The Netherlands, 3 Institute for Advanced Study, University of Amsterdam, Amsterdam, The Netherlands, 4 Instituto de Física 'Gleb Wataghin' & Programa de Pós Graduação em Ecologia, Universidade Estadual de Campinas, Campinas, São Paulo, Brazil, 5 International Centre for Theoretical Physics - South American Institute for Fundamental Research (ICTP-SAIFR), São Paulo, São Paulo, Brazil, 6 Computational Science Lab, Informatics Institute, University of Amsterdam, Amsterdam, The Netherlands

* v.v.vasconcelos@uva.nl

Abstract

Plant and pollinator communities are vital for transnational food chains. Like many natural systems, they are affected by global change: rapidly deteriorating conditions threaten their numbers. Previous theoretical studies identified the potential for community-wide collapse above critical levels of environmental stressors—so-called bifurcation-induced tipping points. Fortunately, even as conditions deteriorate, individuals have some adaptive capacity, potentially increasing the boundary for a safe operating space where changes in ecological processes are reversible. Our study considers this adaptive capacity of pollinators to resource availability and identifies a new threat to disturbed pollinator communities. We model the adaptive foraging of pollinators in changing environments. Pollinator's adaptive foraging alters the dynamical responses of species, to the advantage of some-typically generalists-and the disadvantage of others, with systematic non-linear and non-monotonic effects on the abundance of particular species. We show that, in addition to the extent of environmental stress, the pace of change of environmental stress can also lead to the early collapse of both adaptive and nonadaptive pollinator communities. Specifically, perturbed communities exhibit rate-induced tipping points at stress levels within the safe boundary defined for constant stressors. With adaptive foraging, tipping is a more asynchronous collapse of species compared to nonadaptive pollinator communities, meaning that not all pollinator species reach a tipping event simultaneously. These results suggest that it is essential to consider the adaptive capacity of pollinator communities for monitoring and conservation. Both the extent and the rate of stress change relative to the ability of communities to recover are critical environmental boundaries.

Author summary

Plant and pollinator communities, which support global food chains, are threatened. A significant problem is the reduction of pollination, where not enough plants are pollinated

Fosters Gradual Tipping under Resource Competition and Rapid Environmental Change (Zenodo). Zenodo. https://zenodo.org/doi/10.5281/ zenodo.8268437.

Funding: ST acknowledges the support of the Institute of Advanced Study (University of Amsterdam, IAS excellence students 2022) where part of the work was carried out. FMDM acknowledges the Coordenação de Aperfeiçoamento de Pessoal de Nível Superior Brazil (Finance Code 001). VVV declares financial support of the CSL at IVI (Computational Science Lab at Informatics Institute at the University of Amsterdam, Tenure Track). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

and are at risk of becoming extinct. Environmental change, such as climate change or increasing pesticide use, can cause this reduction. As long as these changes stay under certain boundaries, undesirable ecological processes like the reduction in pollination are reversible. Crossing these boundaries means tipping of pollinator communities might occur; the system might collapse. Pollinator communities can adapt, for instance, through adaptive foraging by changing which plants they prefer in response to environmental deterioration. We include adaptive foraging in a theoretical model of plant and pollinator communities. We show that plant and pollinator communities are vulnerable to the extent of stressors but also to how fast these stressors increase. Our model shows that, with adaptation, the extinction of species is more sequential and spreads out in time and may even temporarily favor especially generalist species.

Introduction

Pollinator communities, comprising interconnected plant and pollinator species, are vital for many ecosystems and enhance their biodiversity [1]. Interactions at the level of these communities ensure the circulation of nutrients. Pollinators pollinate an estimated 87.5% of all existing flowering plant species [2], some of which are the only ones supplying certain vital nutrients to humans [3, 4]. Furthermore, the food industry is highly dependent on flowering plant species, with approximately 75% of crop species and 35% of crop volume relying on pollinators [5]. These numbers show the importance of the mutualistic relationship between plants and pollinators to humans and the biosphere.

Plants provide food to pollinators and pollinators promote pollen grains transference between individual plants, but the exact causal mechanisms that ensure the persistence of pollinator communities remain unclear. Still, one can represent how species interact with each other using bipartite networks, where one set of nodes represents pollinator species and the other set of nodes plant species. Typically, these networks that describe mutualistic interactions consist of a core of generalist species—species that have numerous interactions, many of which they share with other generalists [6, 7]—and a peripheral group of specialists—species with only one or two interaction partners— mostly attached to the generalists. This structure is called a nested network [6, 8, 9]. Nested networks are highly asymmetric since specialists primarily depend on generalists, whereas generalists are less dependent on each of their specialists [6, 7]. Nestedness can help rare specialist species persist [6].

Understanding how pollinator communities remain healthy is crucial, given the declining populations of many species [1, 3, 10, 11]. Multiple environmental stressors contribute to this decline in pollinators. The IPBES report on pollinator trends specifies land-use change, intensive agricultural management and pesticide use, environmental pollution, invasive alien species, pathogens, and climate change as possible drivers of pollinator decline [12]. A panel of 20 experts in the field of pollinator studies ranked the most important drivers of decline globally, in decreasing order of impact, as land cover and configuration, land management, pesticide use, climate change, pests and pathogens, pollinator management, invasive alien species, and genetically modified organisms [13].

Some of these drivers of decline have changed rapidly over the last decades. For instance, rapid climate change has likely been decreasing pollinator diversity in the Northern Hemisphere, leading to a loss of resilience of these pollinators and their ecosystems [14]. Furthermore, a changing climate can cause plants and pollinators to shift out of their climate range or mismatch their phenology and active hours, leading to the decline and possible extinction of

species [3]. Moreover, climate change can lead to species migration and the invasion of other pollinator communities, which can lead to a loss of diversity in those areas [3, 7].

The impact of external stressors on pollinator communities may have far-reaching and non-obvious consequences on ecosystems. For instance, nonlinear dynamics, which can be captured in models of pollinator communities, can present tipping points. Tipping points are defined as sudden, abrupt changes in a system due to a small change in a control parameter [15–20]. This control parameter induces a transition from one state of the system where the community is prosperous to a depauperate one at some critical threshold. Planetary boundaries are the ultimate example of these thresholds [21] above which there is a chance of long-term devastating consequences such as large, irreversible sea level rise [22].

So far, there is no direct empirical evidence for tipping in pollinator communities [23]. In their review, [23] found just one example of tipping in a bee colony [23, 24]. However, some of the theoretical work on pollinator communities has shown that tipping points are possible, especially in models based on Lotka-Volterra type of dynamics [23, 25-29]. Various theoretical works tested the existence of different equilibrium states of the communities for different but fixed values of the control parameters (e.g., interaction strength or driver of decline) [25–28]. Results show that higher drivers of decline cause community collapse [26]. This type of tipping is formally associated with the bifurcation of the equilibrium solutions. Bifurcation-induced tipping is caused by reaching a threshold value of a control parameter—here, the magnitude of the driver of decline—and that critical value is traditionally seen as the "tipping point". However, since these works investigated equilibrium solutions at each subsequent value of the driver of decline (quasi-static analysis), they did not consider the complex time-dependent dynamics that can arise by varying a parameter over time [30]. The equilibrium state might actually be a moving quasi-static attractor, making it possible for rate-induced tipping to exist [31]. Rate-induced tipping occurs due to a rate of change in the control parameter instead of due to reaching a threshold value of the parameter. Conversely, what matters in bifurcationinduced tipping is the absolute value of the parameter, not how slowly or rapidly that value was reached. By ignoring that control parameters are time-dependent, changes in dynamics due to the rate of change are overlooked [19], potentially overestimating the resilience of such systems. Further, it disassociates the critical values of external environments (tipping points) from any particular point in time. Nevertheless, the possibility and risk of rate-induced tipping through rate dependency have so far been ignored in the study of pollinator communities.

On the positive side, the ability to adapt could give pollinator communities greater resilience and, thus, is expected to play an important role in rate-induced tipping. Adaptation, however, is not instantaneous. For adaptation to be an effective coping mechanism, it must occur on a timescale similar to the pace at which the environment changes. Therefore, environmental change should be seen in light of the rate of adaptation. Rate dependence and adaptation have been underappreciated in previous analyses of tipping points in pollinator communities, even though they would bring models closer to the dynamics observed in nature. When the environment changes rapidly, plasticity and evolution are important mechanisms for adaptability.

There is evidence that pollinators show behavioral plasticity in their use of resources and employ various strategies, such as changing the foraging intensity of different species (adaptive foraging) and foraging different species altogether (rewiring). Pollinator communities experience high species turnover and fluctuating abundances [32]. This results in high interaction variability, which can be modeled by the opportunistic attachment of pollinators to plants [33], potentially forming novel or reinforcing existing interactions. By changing their connections, species can adapt quickly to changes in their environment [34]. A sign that this is happening is a high turnover rate of interaction between species [35]. However, this stems mainly

from the variability in the number of active links per species rather than an actual rewiring between species (i.e., forming connections between species that were not previously connected) [35]. Here, we represent these mechanisms in a new model of adaptive foraging.

With adaptive foraging, pollinator species can adapt their investment in plant species to which they can connect, thereby altering the network weights. This strategy is based on the availability of plant resources, which depends on plant abundance and competition between pollinators. Adaptive networks seem to improve the persistence of pollinator communities, and adaptive foraging sustains biodiversity in pollinator communities [36].

As environments keep changing, adaptation may play an increasingly important role. Therefore, including adaptation and dependencies on the rate of change of environmental stressors in current models of pollinator communities is necessary to investigate whether rateinduced tipping is present and how it happens. In this work, the questions are whether i) in adaptive pollinator communities, irreversible tipping can occur due to the speed of environmental change, not just due to the extent of environmental change, and ii) how adaptive foraging changes the overall tipping dynamics of the model. The focus is on adaptive mechanisms at the ecological timescale, not the evolutionary one, since the latter acts at longer timescales. We will use the term adaptation in this sense. We expand on previous models [26] by adding adaptive foraging depending on resource competition and a rate dependency to the driver of decline.

Model

The model is based on previously used mutualistic Lotka-Volterra type models describing pollinator communities (e.g., [26, 27, 29]), extended to include implicit resource competition and adaptive foraging. Fig 1A illustrates the causal mechanisms considered in our model.

Abundance changes

The abundance of each plant (P_i) and pollinator species (A_j) is modeled considering three main components: intrinsic growth rate, mutualistic growth, and intraguild competition (see Fig 1), each contributing to net population growth. These components are described below and are reflected in the following coupled model of plant species indexed by *i*, in a set of S^P plant species, and pollinator species indexed by *j*, in a set of S^A pollinator species,

$$\frac{dP_{i}}{dt} = P_{i}\left(r_{i}^{P} - \sum_{i'=1}^{S^{P}} C_{ii'}^{P} P_{i'} + H\left(\sum_{j'=1}^{S^{A}} \alpha_{ij'} \beta_{ij'}^{P} A_{j'}\right)\right) + \mu,$$

$$\frac{dA_{j}}{dt} = A_{j}\left(r_{j}^{A} - d_{A}(t) - \sum_{j'=1}^{S^{A}} C_{jj'}^{A} A_{j'} + H\left(\sum_{i'=1}^{S^{P}} \alpha_{i'j} \beta_{i'j}^{A} \phi_{i'}\right)\right) + \mu.$$
(1)

Intrinsic growth rate. Each species has a baseline, fixed intrinsic growth rate given by r_i^p and r_i^A for plant and pollinator species, respectively.

Driver of decline. The total growth rate of pollinator species is negatively affected by the driver of decline $d_A(t)$, representing the total environmental deterioration. This driver of decline is either kept fixed or is time-dependent: $d_A(t) = \lambda \cdot t$ with λ the rate of change.

Intraguild competition. The intraguild competition, where each species competes with all species in its own group (i.e., plant species compete with plant species, pollinator species), is a nonlinear term depending on the abundance of the



Fig 1. Dynamics of pollinator communities under external stressors. (A) Causal dynamics of the adaptive foraging model. Plants and pollinators boost each other's abundance through mutualistic interactions; pollinators receive resources from plants and plants are pollinated by pollinators. Interactions are specified by a bipartite plant-pollinator nested network. The strength of the interactions changes over time through adaptive foraging. Pollinators adapt their investment in connected plant species based on the supply-demand ratio of plant resources. Besides, species experience intraguild competition and an intrinsic growth rate. The system is subjected to a driver of decline (d_A)—an external stressor that negatively influences the intrinsic growth rate of all pollinator species. (B) The mutualistic network has a nested structure with a few highly connected species (generalists) and many sparsely connected species (specialists). The edge colors show the weights of the adaptive foraging matrix α , denoting the investment of each pollinator species in plant species. (C) Hysteretic impact of driver of decline d_A (without adaptive foraging). At a critical value of the driver of decline, for any increase of the driver of decline, the pollinator species go extinct (very low abundance). By lowering the driver of decline, the system has another bifurcation-induced tipping point, where, for any decrease of the driver of decline, it recovers at the point of recovery where the first species are reintroduced.

species involved and competition matrices $C_{ii'}^p$ and $C_{jj'}^A$ for plant and pollinator species, respectively.

Mutualistic interactions with resource competition. A Holling type-II function gives the mutualism-derived growth, $H(\rho)$, which ensures the saturation of the mutualistic relation as the overall mutualistic benefit, ρ , increases. Precisely, $H(\rho) \equiv \rho (1 + \rho/h)^{-1}$, with *h* as a

horizontal asymptote with a different value for each species [37]. The mutualistic benefits, ρ , are the sums passed to the *H* functions for pollinators and plants in Eq.1. They are different for pollinators and plants, so we call them ρ_i^A and ρ_i^p , respectively.

To establish the mutualistic benefit for plant and pollinator species, we extend previous models (e.g., [26, 27, 29]) by considering that pollinators can change their strength of interaction with each of their connected plant species. The strength of interaction and the resources available determine the mutualistic benefit the species derive.

Formally, this is captured by the element-wise product of two matrices, an evolving foraging effort α and a static matching matrix β , that, together with resource availability, make up the interaction strength.

The matching matrix β encodes whether a connection between a plant and pollinator species can exist and its innate efficiency. The innate efficiency describes how well two species can interact with each other given their phenotype, the accessibility of the plant to the pollinator, and other unknown constraints [38]. Matrix β also contains forbidden links (FL), connections that cannot exist due to phenological uncoupling [38]. We assume that evolutionary processes do not play a role at the timescales we consider, and no species can rewire their interactions completely; therefore, β is fixed.

Whether two species can interact with each other (and, thus, whether the β -value between two species is nonzero) is dictated by a bipartite network describing the connections. We generate these bipartite networks using an algorithm as described in Section B of <u>S1 Text</u>. The algorithm takes as input the number of plant species, number of pollinator species, nestedness, fraction of forbidden links, and connectivity. These networks have a similar—nested—structure as real-life plant-pollinator networks [8, 26].

The foraging effort, α , is the relative investment a pollinator species *j* makes in a certain plant species *i*. It encompasses the effort—the amount of resources and time—that pollinators decide to invest in each plant. The pollinators, not the plants, are, therefore, leading in the choice of mutualistic interaction strengths. Plants are assumed to have no selective mechanism apart from the matching β and their resource abundance. High matching and high abundance make plants more attractive to pollinators. The total foraging effort is constrained by

 $\sum_{i}^{s^{r}} \alpha_{ij} = 1$, meaning that each pollinator species *j* divides a total foraging effort of 1 across all connected plant species. The foraging effort α is initialized such that the initial foraging effort is evenly distributed over all connected plant species.

The values for β are sampled uniformly for each connection existing in the bipartite network and subsequently normalized by the number of connections a plant (K_i^p) or pollinator (K_j^A) species has. As a second step, the resulting β values are divided by the initial foraging effort (α) values such that the product of α and β results in values normalized according to the degree of a species:

$$egin{aligned} η_{ij}^{P} \Rightarrow & rac{eta_{ij}^{P}}{(K_{i}^{P})^{\eta} lpha_{ij}^{\min}} \;, \ η_{ij}^{A} \Rightarrow & rac{eta_{ij}^{A}}{(K_{i}^{A})^{\eta} lpha_{ij}^{\min}} \;. \end{aligned}$$

The strength at which the network topology matters for the mutualistic interaction strength is controlled by the exponent η . A higher η rewards specialization over generalization, and vice versa [26]. We use a value of 0.5 for all our simulations.

The resource availability is modeled based on the amount of plant resource (food) available and the demand for this resource. It is directly coupled to plant abundance and is not a state variable on its own. The availability is represented by the supply-demand ratio $\phi_i \equiv \frac{\text{supply}}{\text{demand}}$. Pollinators compete over the resource capacity of each plant species. We assume that the plant species P_i has a supply equal to its abundance: supply = P_i .

The demand is given by the amount of resources that the connected pollinator species aim to extract. As previously stated, per capita, pollinator species *j* can extract $\alpha_{ij}\beta_{ij}$ from the resource. The total demand of pollinator *j* over plant *i* is this multiplied by the abundance: $\alpha_{ij}\beta_{ij}A_j$. The total demand for resources of plant species *i* is given by the sum over all connected pollinators: demand = $\sum_{j=1}^{S^A} \alpha_{ij}\beta_{ij}^AA_j$.

Finally, we define a parameter q—called resource congestion [39]—that models the excludability of the resource. Specifically, the resource congestion, q, models the strength of the competition over resources by altering the demand such that

$$\phi_i \equiv \frac{\text{supply}_i}{\text{demand}_i} = \frac{P_i}{\left(\sum_{i=1}^{S^A} \alpha_{ii} \beta_{ii}^A A_i\right)^q},$$
(2)

with $0 \le q \le 1$. At q = 0, Eq.2 equals P_i , meaning that there is no resource competition. For q > 0 there is competition over resources.

The mutualistic benefit for the pollinator species then becomes

$$ho_j^{\scriptscriptstyle A} = \sum_{i=1}^{S^p} lpha_{ij} eta_{ij}^{\scriptscriptstyle A} \phi_i \, ,$$

and for the plant species,

$$ho_i^{\scriptscriptstyle P} = \sum_{j=1}^{S^A} lpha_{ij} eta_{ij}^{\scriptscriptstyle P} A_j$$

Migration. Both pollinator and plant species have a positive migration rate μ to allow extinct species to regain abundance, giving rise to the full set of equations for the model (Eq 1). Migration μ is considered small so as not to influence the dynamics of the model (namely, maintaining a stable low abundance state for all species). This assumption is validated through a sensitivity analysis (see SI5).

Adaptive foraging

The pollinators, rather than the plants, lead the choice of mutualistic interaction strengths. We assume that pollinators have a selective mechanism based on their inherent matching with plants and the resource abundance of plants. High matching and high abundance make plants more attractive to pollinators. Pollinators use adaptive foraging by changing the weights of their connections to plants to increase their gathered resources (represented in different shades in Fig 1B).

Since each pollinator seeks to maximize its gathered resources, we assume a local optimization function for α_{ij} . We use a replicator equation (similar to [36]), given in its general form by

$$\frac{d\alpha_{ij}}{dt} = \alpha_{ij} \Big(f_{ij} - \langle f \rangle_j \Big) = \alpha_{ij} \left(f_{ij} - \sum_{i'=1}^{S^{\nu}} \alpha_{i'j} f_{i'j} \right),$$

with f_{ij} denoting the fitness of the interaction between species *i* and *j*. The fitness here is defined as $f_{ij} \equiv \beta_{ij}^A \phi_i$. The strength of the adaptation is controlled by a parameter ν ($0 \le \nu \le 1$).

This parameter generates a trade-off between keeping the original weights in the α matrix ($\nu = 1$) and purely optimizing the foraging effort for the pollinators ($\nu = 0$),

$$\frac{d\alpha_{ij}}{dt} = \alpha_{ij} \left(\beta_{ij}^A \phi_i - \sum_{i'=1}^{S^P} \alpha_{i'j} \beta_{i'j}^A \phi_{i'} \right) (1-\nu) + \left(\frac{1}{S_j^P} - \alpha_{ij} \right) \nu,$$
(3)

where S_j^p is the number of plant species to which pollinator species *j* is connected. Adaptive foraging decreases with increasing *v*, and there is no adaptive foraging at v = 1. The model is the same as described by [26] when setting q = 0 and v = 1 (i.e., no adaptive foraging and no resource competition).

The use of the replicator function is convenient since it guarantees the conservation of the total available foraging effort per pollinator species, as Eq.3 conserves the quantity $\sum_{i}^{S^{P}} \alpha_{ij} = 1$ if initialized as such. In addition, the use of replicator dynamics can also be seen as a selection process that occurs within each species for individuals with different plant preferences.

The model just described sets the foundation for a series of simulations. Next, in the Methods section, we delve into the specifics of how we operationalized these, detailing the parameters and scenarios under which the simulations were conducted.

Methods

The baseline parameters and ranges for the simulations are given in Table A in <u>S1 Text</u>. The model was solved using the LSODA integrator of SCIPY with $RTOL = 10^{-4}$ and $ATOL = 10^{-7}$ [40].

Feasibility of networks

Different interactions among species generate different communities. To ensure that each mutualistic network had similar characteristics, we only investigated networks that had what we call a "feasible solution", meaning that all their species survive in equilibrium when there is no external stress ($d_A = 0$) [26]. To this end, parameters were sampled from Table A in S1 Text, and the model ran to equilibrium. If all species were alive at equilibrium, the parameters were accepted. If not, the parameters were resampled up to 100 times. If, after 100 tries, no feasible parameters were found, a new network was generated. We considered species extinct when their abundance is below 0.01 [26]. The fraction of feasible networks can be seen as a measure of the persistence of the diversity of a community.

Bifurcation-induced tipping

Bifurcation-induced tipping points were found by calculating the equilibrium trajectories of species abundance depending on the direction of change of the driver of decline d_A . The abundances of all species were set to initialize at 1 and evolved until equilibrium was reached for $d_A = 0$. The driver of decline d_A was increased in steps of 0.02, each time running the simulation until it reached equilibrium. This continued until all pollinator species were extinct. The value of d_A when the last species went extinct was denoted by the point of collapse $d_A^{collapse}$. From this extinct state, the simulation was repeated, but now the driver of decline d_A was decreased each time in steps of 0.02 until reaching the point of recovery $d_A^{recovery}$: the value of the driver of decline d_A at which the first species returned to existence.

The region of bistability is the region between d_A^{recovery} and d_A^{collapse} for each network, given $d_A^{\text{recovery}} < d_A^{\text{collapse}}$. The inequality condition guarantees that there are two stable equilibria for a given parameter set and network.

To assess the influence of the competition over resources between pollinators on the stable states of the model, the point of final collapse d_A^{collapse} and first recovery d_A^{recovery} of the pollinator species were calculated for different strengths of resource congestion q. For each resource congestion q, the results were averaged across 100 networks. These networks were generated as described above, except that the networks were not resampled if after 100 parameter samples the network was still not feasible, meaning even the non-feasible ones could be accepted. The reason for this is that for some values of the resource congestion not many feasible networks might exist. This also allowed us to use the fraction of feasible networks as an indicator of the persistence of the pollinator communities under a given resource congestion q.

Rate-induced tipping

To identify rate-induced tipping, we checked if all pollinators went extinct at a certain critical rate of change of the driver of decline before bifurcation-induced tipping occurred. Since bifurcation-induced tipping occurred at the point of collapse d_A^{collapse} , the simulation was run for different rates of change λ until a maximum value of d_A , denoted by the fraction θ of the point of collapse d_A^{collapse} ($0 \le \theta \le 1$), was reached. By varying θ , we could check whether the tipping response was purely due to the rate of change λ (in which case we should see rate-induced tipping for most values of d_A^{max}) or due to a combination of the rate of change and bifurcation-induced tipping (in which case we should only see rate-induced tipping for a d_A^{max} close to d_A^{collapse}). Each simulation was averaged across 100 feasible networks. The number of pollinators alive at the end of the simulation was expressed in terms of the relative persistence of the pollinator species. This is the fraction of pollinator species alive for a given rate of change λ , compared to the number of species alive at the lowest rate of change of λ investigated. Therefore, it measures the change in the number of species alive due to increasing rates of change λ .

To test the influence of the initial abundance on rate-induced tipping, the value of the driver of decline d_A at which all pollinators went extinct was calculated for different rates of change λ of the driver of decline. To avoid confusion with the bifurcation-induced tipping point d_A^{collapse} , we call this point the point of extinction d_A^{extinct} . However, both indicate the value of the driver of decline d_A that all pollinator species go extinct. Simulations were carried out for two different initial conditions: low abundance for all species ($S^{\text{init}} = 0.1$) and high abundance for all species ($S^{\text{init}} = 2$). The low abundance condition was chosen such that the abundance was in the bistable range of most pollinator species, while the high abundance condition was chosen such that most pollinator species were around or above the upper stable state (the abundant state). Each simulation was started from $d_A = 0$ with a rate λ until all species were extinct at the point of collapse d_A^{collapse} , averaged across 100 feasible networks.

We used these methods to test the predictions of the model under various conditions. By systematically altering key parameters and observing the results, we aim to unravel the complex dynamics of pollinator communities in response to environmental stressors. The following Results section presents the outcomes of these simulations, shedding light on the practical implications of our model.

Results

Perturbed pollinator communities can exhibit rate-induced tipping with and without adaptation

We tested the behavior of the pollinator communities in response to external environmental change, considering both adaptive and nonadaptive populations with different resource congestion. If the environment is changing very slowly without adaptation, after a critical value of

the driver of decline, d_A^{collapse} , the pollinator community collapses (see Fig 1C). For perturbed (low initial abundance) communities without adaptive foraging, tipping occurs at specific rates, i.e., rate-induced tipping, when the driver is close to, but below, the point of collapse d_A^{collapse} when initial abundances are low (Fig 2). Thus, this tipping point depends on both the magnitude and the rate of change of the driver of decline. The collapse of pollinator communities happens only around a maximum value of the point of collapse d_A^{collapse} of around 70% or more (Fig 2B). In the supplementary material, Fig F in S1 Text shows the full bimodal distribution of pollinator persistence over the networks for non-adaptive communities, different rates, and resource competition.

At low rates of change, λ , the species persistence is mainly defined by the maximum driver of decline d_A^{max} . Since the rate of change is slow, the system is close to equilibrium most of the time. For low rates of change and a d_A^{max} below the point of collapse d_A^{collapse} , this means no tipping occurs. At higher rates of change λ , the species persistence depends more on λ . In this case, all species go extinct if and only if the maximum value of the driver of decline is close enough to d_A^{collapse} . This is the case for almost all networks for a d_A^{max} of 80% of the point of collapse d_A^{collapse} .



Fig 2. Species persistence collapses for high rates of environmental change within environmental ranges that are otherwise presumed safe. (A) Three scenarios are represented for communities without adaptive foraging, perturbed to start with a low initial species abundance (out of equilibrium). (Inset) Stress in communities increases over time as the driver of decline increases at different rates, λ , up to a maximum value, $\theta \cdot d_A^{collapse}$. The black line represents the point of collapse above which a fixed value of the driver of decline leads to the collapse of all communities and below which some communities are sustained. The maximum value of the driver of decline up to 90% of the critical value, $d_A^{collapse}$, dot-dashed green line an increase up to 50%, and dashed pink an increase up to 20%. In this panel, species persistence is calculated as the fraction of pollinator species alive relative to the number of species alive at the lowest rate of change measured ($\lambda_{min} = 10^{-4}$). (B) The persistence of species decreases as a function of the maximum value of the driver of decline, represented as a fraction θ of the point of collapse $d_A^{collapse}$ to collapse decreases to a value close to, but lower than, the point of collapse $d_A^{collapse}$ at a fast enough rate. In this panel, species persistence is calculated as the fraction of pollinator species alive at $\theta =$ 0 (no external stressor). Initial species abundance $S^{init} = 0.1$ for all simulations. The results are averaged over 100 feasible networks, for which all 15 plant and 35 pollinator species survive under no stress, with the bands representing the first to third quartile ranges. Other parameters in Table A in S1 Text.

https://doi.org/10.1371/journal.pcbi.1011762.g002

Adding adaptive foraging to the model changes the response to the rate of change, but rateinduced tipping is still present (Fig 3). For low magnitudes of the maximum driver of decline, adaptive communities already experience partial, and sometimes even full, extinction. There is a decrease in relative pollinator persistence over a large region of the fraction θ of the point of collapse d_A^{collapse} (Fig 3B). In comparison, in the nonadaptive case (Fig 2B), there is a more clear threshold around $\theta \approx 0.7$ where pollinator communities go from relatively high persistence to collapse. Adaptive pollinator communities are, thus, more likely to collapse partially over a range of maximum decline drivers rather than exhibit a complete collapse of all species at a specific value of the driver (see Fig G in <u>S1 Text</u> for the complete bimodal distributions of pollinator persistence over the networks showing partial collapse). This suggests that adaptive pollinator communities might show a more gradual tipping process in which a few species go extinct at a time rather than a single collapse. Later, we show which species are affected throughout the gradual tipping process.

As we have shown in Figs 2 and 3, the rate of change λ in the driver of decline d_A influences the values of the driver of decline d_A at which the last pollinator goes extinct. We refer to the critical value at which this rate-induced tipping occurs by the point of extinction d_A^{extinct} —the value of the driver of decline d_A at which the last pollinator goes extinct. The point of extinction d_A^{extinct} increases mostly monotonically with the rate of change λ for high initial abundances (Fig 4). This is because, for high initial abundances, the species start with an abundance around or above the stable state with high abundance. This means species abundances change slower



Fig 3. Pollinator communities with adaptive foraging still collapse at high rates of change but less abruptly in the extent of environmental change. Figure equivalent to Fig 2, but considering adaptation and resource congestion. Adaptive communities respond to an increasing driver of decline by reweighing their connections. (A) Rate-induced transitions are still present, with some communities exhibiting rate-dependent tipping at 50% of the point of collapse. Non-monotonicity is within the error range, thus, non-significant for the number of simulations. (B) Overall, pollinator persistence is more sensitive to rates of change in a larger domain of changes in the driver of decline, θ , than for communities without adaptive foraging. Some particular networks see an increase in persistence, especially for small changes and low rates of change, leading to distinct relative persistence levels above 1. For all simulations, initial species abundance of S^{init} = 0.1. Adaptation strength of v = 0.7 and resource congestion q = 0.2. The results are averaged over 100 feasible networks, for which all 15 plant and 35 pollinator species survive under no stress, with the bands representing the first to third quartile ranges. Other parameters in Table A in S1 Text.

https://doi.org/10.1371/journal.pcbi.1011762.g003



Fig 4. Higher rates of change in the driver of decline lead to extinction at lower values of the driver of decline only for perturbed communities. The figure shows the value of the driver of decline d_A at which all pollinators go extinct, d_A^{extinct} , as a function of the rate of change λ of the driver of decline. For a low initial pollinator abundance (left panels), after a critical value of the rate of change λ , d_A^{extinct} has a nonlinear response. This effect disappears for higher resource congestion q, while it increases with stronger adaptation. For a high initial pollinator abundance d_A^{extinct} increases monotonically with the rate of change λ . The results are averaged over 100 feasible networks per case (lines), shown with the standard deviation across networks (band). v = 0.7 for the case with adaptive foraging. The initial abundance for all species is $S^{\text{init}} = 0.1$ for the low initial abundance condition and $S^{\text{init}} = 2$ for the high initial abundance condition. Other parameters in Table A in S1 Text.

than the driver of decline d_A . In other words, the system responds slower to forcing than the forcing itself changes. In this case, rate-induced tipping is not present.

On the other hand, at low initial abundance, there is a non-monotonic response of the point of extinction d_A^{extinct} as a function of the rate of change λ . More specifically, inside the bistable region of the system (between the two stable states)—where species have a low initial abundance—higher rates of change λ can lead to a lower d_A^{extinct} after a critical value of λ as shown in Fig 4. This supports the notion that the pollinator communities need to be perturbed for rate-induced tipping to be possible. Furthermore, Fig 4 shows a stronger non-monotonic behavior for pollinator communities with adaptive foraging than without adaptive foraging.

The collapse of pollinator communities can thus not be fully understood by bifurcationinduced tipping or rate-induced tipping separately. First, the system must possess bistability for rate-induced tipping to occur. Without bistability, there is no opportunity to tip from one branch to another since there is only one branch. High rates of change can then, in this case, not force the system to tip from one state to another. Second, the presence of rate-induced tipping depends on the trajectory in state-space. This "tipping trajectory" depends on i) the rate of change of the driver of decline, which needs to be high enough (Figs 2A and 3A), ii) the total magnitude of the driver of decline, which needs to be close enough to the point of collapse as dictated by the equilibrium trajectories (Figs 2B and 3B), and iii) the perturbation applied to the system (e.g., the initial abundance of species), which needs to be below the upper stable branch in the bistable area of the bifurcation diagram (Fig 4).

Influence of adaptation and congestion on stability

To better illustrate how adaptive foraging alters the dynamics under increasing stress, Fig 5 shows a representative simulation of plant and pollinator abundances under increasing drivers of decline d_A . The plant and pollinator species start with a low abundance ($S^{init} = 0.1$). The driver of decline d_A is increased from 0 with a rate of $\lambda = 0.05$. Given the slow increase in the driver of decline d_A , both plant and pollinator species initially recover their abundances (Fig 5A and 5B). Most plant and pollinator species reach their peak around 10 time units after which their abundances decrease. Eventually, all pollinator species go extinct.

During this process, pollinators shift their foraging effort so that it benefits them the most. Panels C-H of Fig 5 show how their foraging effort changes, sorted by their number of connections (degree). After roughly 10 time units, the driver of decline increases to such a level that many species start to deteriorate. At this point, some large changes occur in the foraging effort. Most notably, a few pollinators manage to invest heavily in one or two plant species, thereby temporarily offsetting their decline in abundance. This leads some plant species to stabilize their decline for an extended period, which allows certain high-degree pollinator species to persist significantly longer than other, lower-degree pollinator species. The two pollinator species with degree 9 (Fig 5H) even temporarily benefit from the extinction of other species, as can be observed by a temporary increase in their abundances (Fig 5A). They increase their foraging effort significantly in one of their connected plant species (Fig 5B). At the same time, many low- to mid-degree pollinator species do not manage to adapt their foraging effort in such a way as to offset the negative effect of the driver of decline. Eventually, the pressure of the driver of decline gets so high that all pollinator species go extinct.

This representative simulation also gives an explanation for the curve in Fig 3B where at low θ values the relative pollinator persistence already decreases before reaching 0 between 0.4 $< \theta < 0.6$. As Fig 5A illustrates, some pollinator species go extinct faster than others (faster relates to a lower driver of decline value). In combination with Fig 3, where the simulations ran until equilibrium, it shows that partial pollinator community collapse occurs when there is adaptive foraging.

Both the adaptation strength v and the resource congestion q influence these dynamics and, thus, the bifurcation diagram of the pollinator abundances. Three examples are shown in Fig 6. Without adaptation, both pollinator species collapse and recovery are abrupt as a function of the control parameter driver of decline d_A . Adding resource congestion leads to an almost complete disappearance of bistability in this area due to the disintegration of the network structure. Adding adaptation leads to the reintroduction of bistability. At the same time, both the collapse and the recovery of pollinators become less abrupt. Looking at individual species, as illustrated in S1 Text Figs B-E, some high-degree and highly abundant species can survive longer after the partial collapse of the system. We see a similar effect in the adaptive pollinator community in Fig 5. Some highly abundant, generalist species get a temporary boost after the



Fig 5. The coevolution of foraging effort and species abundances under increasing driver of decline d_A . The top row shows the evolution of pollinator and plant abundances under linearly increasing driver of decline d_A with rate $\lambda = 0.05$, starting from a low abundance condition of $S^{init} = 0.1$, with adaptation strength of v = 0.7 and resource congestion q = 0.2. Each line style and color combination represents a single pollinator species in all graphs (except the plant abundance graph, top right). For example, there is one pollinator species with degree 3. Since the degree is 3, there are three solid blue lines (one for each connection to a plant species). Another example is the two pollinator species with degree 4, thus showing eight lines (four solid lines for one species and four dashed lines for another species). Since the values of the foraging effort α for each individual species add up to 1, the evolution of the foraging effort α is not shown for pollinators with degree one since they have a constant $\alpha = 1$ to their single connected plant species. Pollinators with high degree rapidly become the most abundant. Furthermore, the foraging effort α drastically changes—especially around 10 time units when most species reach their peak abundance. The two pollinator species with degree 9 survive the longest and also have one plant species in which they invest most of their foraging effort after 10 time units. Other parameters in Table A in S1 Text.

partial collapse. This process leads to a more gradual decrease in the average pollinator abundance, as shown in Fig 6.

The resource congestion q (dictating how strong the competition over resources is) determines whether pollinator communities possess hysteresis. As Fig 7 shows, hysteresis is always present in pollinator communities without resource competition. However, increasing resource congestion leads to a loss of the bistable states that define hysteresis in both the model with and without adaptive foraging, hereby merging the point of collapse $d_A^{collapse}$ and the point



Fig 6. Influence of adaptive foraging and resource congestion on the bifurcation diagram. Adaptation changes the bifurcation diagram. The collapse is less abrupt for pollinator communities with adaptive foraging. Resource competition decreases the size of the bistable area if there is no adaptive foraging. S1 Text Figs B-E show the full bifurcation diagrams of all individual pollinator species for different settings of v and q. The results are averaged over 100 networks per parameter setting, with the error bands showing the standard deviation. Parameters in Table A in S1 Text.

of recovery d_A^{recovery} . Bistable states disappear with higher resource congestion q for systems with adaptive foraging compared to systems without adaptive foraging.

The increase of the driver of decline of collapse d_A^{collapse} and recovery d_A^{recovery} is illustrated by the bifurcation diagrams shown in S1 Text Figs B-E. With increasing resource congestion q, some pollinator species have a lower abundance at zero stressors and go extinct faster. In contrast, other species have a slower and more gradual decrease in abundance once the resource congestion q increases. Due to the increased competition for resources, the stress on some species is high, which causes them to go extinct fast. However, the longest-surviving species have little competition over their resources with a high driver of decline d_A , making them less prone to extinction. This is because when there is low competition over resources, there is a high supply-demand ratio, which boosts the growth rate of these species. On the downside, this results in a significant loss of species diversity. In general, the tipping points become less synchronized as resource congestion q increases.

In the case of adaptive foraging, competition over resources helps pollinators spread their foraging effort more evenly by increasing the profits for plants with few interactions. However, once competition over resources increases too much, some plant and pollinator species cannot survive. This makes secondary extinctions possible, which depend on the extinct species. This process leads to more infeasible networks.



Fig 7. Adaptability and resource congestion affect hysteretic patterns and the viability of plant-pollinator networks. For adaptive pollinators, intermediate levels of resource congestion increase the overall persistence of ecological networks. The point of collapse and recovery of pollinator species increases as a function of resource congestion *q* without (A) and with adaptive foraging (B). For low resource congestion, the system possesses bistable states which disappear after a critical value of the resource congestion. Resource congestion also affects the feasibility of the networks— networks for which all 15 plant and 35 pollinator species survive under no stress. An intermediate level of resource congestion is required for the adaptive model to produce feasible networks. The orange arrows indicate the resource congestion strength *q* chosen for the simulation of Figs 2 and 3. These values were chosen such that the systems possess bistable states—as observed in the non-overlapping points of collapse and recovery—and have a high fraction of feasibility. For low resource congestion, adaptability increases the range of drivers of decline at which pollinator communities do not collapse, increasing resilience in the Holling sense [41]. (A) v = 1 and (B) v = 0.7. The results are averaged over 100 networks per value of resource congestion *q* with the error bars showing the standard deviation. Other parameters in Table A in S1 Text.

Influence of adaptation strength

The feasibility of the networks also depends greatly on the strength of adaptation v. At higher adaptation, fewer species survive because a few plant species gather high foraging effort from pollinators, boosting these plants' abundance. Thus, more pollinators pollinate these highly abundant plant species and lose interest in other plant species, leading to their extinction. This, in turn, causes the extinction of pollinator species that depend on these now-extinct plants. When there is a high level of adaptation and only a few generalists with high abundances, only the pollinators connected to them survive. Although this could lead to a strong increase in abundance in these selected plant and pollinator species, the structure of the network is also highly affected. Due to the extinction of many specialist plants, the network structure disintegrates. This leads to the disappearance of highly synchronized extinction events (this is apparent in Fig D in S1 Text: at high adaptation, there are many extinction events of just a few pollinators per time, as opposed to the more synchronous tipping in Fig A in S1 Text). Due to both the higher individual abundances of successful species and less synchronized tipping, a higher driver of decline d_A is necessary to kill all pollinator species. At the same time, pollinators not connected to these generalists either do not survive under zero external stressors-leading to infeasible networks-or already go extinct at low values of the driver of decline d_A .

A Sobol sensitivity analysis in SI5 shows the model parameters' influence on the networks' feasibility and the point of collapse. The resource congestion q and adaptation strength v have the strongest influence in both cases. Furthermore, the migration rate μ influences the results very little, thereby validating our assumption that this parameter does not significantly influence the dynamics of the model.

The above results provide a detailed account of how pollinator communities react under different stress scenarios, highlighting the critical role of adaptation and the rate of environmental changes. In the Discussion, we interpret these findings in a broader context, discussing their implications for ecological theory, pollinator conservation strategies, and future research directions.

Discussion

The species that form pollinator communities are codependent and interact in complex ways. This makes it challenging to study how the rate of change of external environmental forces affects the stability of a community. Some of the existing models do not capture adaptive processes and, therefore, are inadequate to understand how changing environments affect pollinator communities. Here, we provide a model that includes adaptation and, thereby, the capacity to respond to change, which can be used to understand the future state of pollinator communities more accurately.

With this model, we shed light on overlooked perils, such as rate-induced tipping. We demonstrate that rate-induced tipping is present in a theoretical model excluding and including adaptive foraging in pollinator communities. Rate-induced tipping points, where all pollinators go extinct, can occur for values of the driver of decline whose effects are otherwise considered reversible when the driver of decline is constant or slowly varying. For rate-induced tipping to occur, the system needs to possess bistable states and be perturbed out of equilibrium.

The existence of bistability in pollinator communities is strongly influenced by both competition over resources and adaptive foraging, highlighting the importance of including natural mechanisms in models to anticipate the responses of real ecosystems. Here, we show that high resource competition—through high resource congestion—can lead to the disappearance of hysteresis. In other words, there is only one stable state for each value of the driver of decline, whereby tipping is no longer present. On the other hand, in the case of adaptive foraging, resource competition is vital to create feasible networks. Without resource competition, only a small fraction of networks are feasible in adaptive pollinator communities.

Importantly, the adaptive capacity of pollinator communities influences rate-induced tipping. Systems possessing adaptive foraging have a stronger response to different rates of change of the driver of decline such that some pollinator species already go extinct at slow rates. This is because rapid adaptation can temporarily mask the negative effects of an increasing driver of decline, but it can also lead to additional competition for resources, which can negatively affect some species by reducing their abundance.

The evidence for community-scale collapse draws mainly from theoretical work, without direct empirical observations of such an event [23]. This lack of empirical observation could be due to the rarity of such an event. Indeed, our work supports the idea that community-scale collapse—in its traditional narrow sense—is rare because species in adaptive pollinator communities are unlikely to disappear simultaneously. We were able to gain this insight with our model for two reasons. First, as opposed to previous studies that looked at static external environments, the two timescales (that of the external environment and that of the pollinators) are coupled. Second, by including adaptive foraging, we include part of the natural adaptive capacity of pollinator communities to react to environmental change.

Notably, here we speak of rate-induced tipping and bistability as community concepts, showing how the critical values of the rate of change of driver of decline can differ between pollinator communities. However, when extended to the species scale, this region is different for each species, with different species collapsing under different conditions. The different

species are heterogeneous, have asymmetric interactions, and, consequently, do not all react equally to environmental degradation. What we observe in our model with adaptive foraging is a sequence of "failures"—tipping events of single or small groups of pollinator species before the whole community disappears. Some species abundances—mainly generalist species —even see a temporary boost after the extinction of a few species due to the decrease in competition over resources. Therefore, generalists are more robust to rapid environmental changes than specialists. The fact that specialist species are the first to disappear seems to align with empirical studies [10, 23, 42].

The new model more accurately represents a variety of mechanisms present in real pollinator communities, such as added competition over resources through the supply-demand ratio and adaptive foraging. Our study can capture adaptation on ecological timescales by including adaptive foraging through resource competition and explicit environmental change. Nevertheless, there are other adaptive mechanisms in pollinator communities besides adaptive foraging that we did not investigate. Two of them are (co-)evolution and the rewiring of species.

Evidence shows that co-evolution occurs between plant and pollinator species [43]. Plants can adapt through evolution by genetic variation [44], being able to do this quickly when pollinator communities change suddenly. Co-evolution leads to different networks [45]. In ecosystem models in general, adaptive responses at the genetic level can play an important role in evaluating the response to changing external stressors [46]. Evolution may be vital on larger timescales when the environment changes slowly enough for evolutionary mechanisms to activate or when evolution is fast [47–49].

Moreover, contrary to our assumption of a fixed set of possible interactions between plant and pollinator species, these species can also form novel interactions through rewiring by forming new connections with existing or immigrant species [32]. This contrasts with our definition of adaptive foraging, where only the strength of interactions changes, but no new connections are possible. Rewiring occurs especially from pollinators to plants that have high abundances and when the trait matching is increased between them [34]. Rewiring of the species can further increase resilience [50]. It can also help pollinator communities adapt to strong phenological changes [10]. Furthermore, species are more resistant to extinction due to the rewiring of species interactions in mutualistic networks [51]. Yet, the effect of rewiring on rate-induced tipping is still unknown. Without this knowledge, we might have underestimated the resilience of pollinator communities and, at the same time, found a smaller role for adaptation in rate-induced tipping. Therefore, the conclusions of this study stem from a pessimistic scenario but are nevertheless important for calibrating preventive measures so that pollinator communities remain far from extinction.

Including all these effects to add realism comes with the drawback that models with higher effective dimensions tend to produce more uncertain estimates, especially when data are scarce [52]. Thus, a parallel effort of mechanism calibration is necessary.

We show that including sufficient natural mechanisms in models describing pollinator communities changes the resilience of these systems, opening perspectives for the management of real ecosystems. Introducing resource competition leads to two different regimes: one in which hysteresis is present and one in which it is not. Although the highest rate of feasible ecological networks is found in systems with hysteresis, based on our results, it cannot be ruled out that there are pollinator communities without hysteresis. This should be considered when estimating the effort needed to restore pollinator communities from collapse. Restoring systems without hysteresis is easier than restoring systems with hysteresis since the former is reversible while the latter is not.

So far, ecosystem management has been concerned with establishing limits to species abundance and limits to the external environmental conditions of these ecosystems. For instance, the group on the post-2020 Global Biodiversity Framework mentions the goal of a "reduction of ecological footprint in an equitable manner (. . .) within planetary boundaries by 2050" [53]; other works advocate the use of fixed targets as opposed to net rates [54]. The work on rate-induced transitions, to which our study contributes, shows that to determine targets of safe operation, it is important to consider not only the extent but also the rate of change in external conditions. Only then can we make a correct assessment of the resilience and recovery of these ecosystems. By neglecting that external conditions change over time, alterations in dynamics due to the rate of change are overlooked [19]. In practice, the rate of change can be a boundary in itself.

Since we showed that perturbed pollinator communities can experience rate-induced tipping, rate-induced tipping might be expected under stochastic forcing [27]. The next question to be explored is what the risk is for rate-induced tipping under stochastic forcing.

To further test the disappearance of the bistability due to resource competition, it would be interesting to examine existing models that include resource dynamics, e.g., [36] and [55], for the presence of hysteresis as a function of induced mortality through added drivers of decline. These models, with higher complexity, have explicit resource dynamics and some type of adaptive foraging, instead of the implicit parameterized resource dynamics used in our study. The conditions for the presence or absence of hysteresis in these models with more fundamental processes would make it clearer when pollinator communities possess bistable states and whether they experience rate-induced tipping.

Beyond further extensions, further analysis of this model could also provide additional insights into the resilience of pollinator communities. In our model, some specific networks saw an increase in relative pollinator persistence with environmental change. Systematically identifying the properties of these networks could also contribute to a better understanding of what structures improve pollinator communities resilience. Additionally, the analysis of possible different extinction pathways and consequent multi-stability could also shed light on the path dependency of these communities.

Finally, our work also has implications for the study and notions of resilience. Since Holling [41], the concept has evolved from local bounce-back speed metrics to the ability of systems to reorganize under change and find new (but productive) equilibria. In the context of pollinator communities, these concepts are useful in defining resilience in terms of determining the region of external conditions for plants and pollinators that allow the system to perform such a transformation [56]. However, these external conditions cannot be seen (quasi-)statically. New uses of resilience must be sure to incorporate rapid external change.

Supporting information

S1 Text. Text with supporting information. The text with supporting information contains additional computational experiments and algorithms. It is divided into five sections. **Section A. Simulation set-up**. Includes: **Table A. Default model parameters**. The default parameter values and ranges used in all simulations, unless otherwise specified. AF = Adaptive Foraging, $\sim U(\cdot) = drawn$ from a uniform distribution at the beginning of each simulation. **Section B. Network generation**. Describes the network generation algorithm and contains: **Fig A. Adjacency matrix of a nested network and forbidden links** Adjacency matrix of a pollinator network on the left with the corresponding forbidden links matrix on the right. Black squares denote the presence of a link. The connectance is 0.15 and the fraction of forbidden links is 0.3. There is a clear difference visible between generalist species and specialist species, in the sense that there are a few species with high connectivity and many with low connectivity. **Section C. Dependence of hysteresis on resource congestion and adaptation**. Contains various

additional computation experiments. Fig B. Hysteresis for increasing resource congestion q for the non-adaptive model. Equilibrium abundance of pollinator species as a function of the drivers of decline d_A for increasing resource congestion q for the non-adaptive model. The blue lines show the equilibrium trajectory for increasing d_A and the orange lines show the equilibrium trajectory for decreasing d_A . See Table A for the parameters used. Fig C. Hysteresis for increasing resource congestion q for the adaptive model with v = 0.8. Equilibrium abundance of pollinator species as a function of the drivers of decline d_A for increasing resource congestion q for the adaptive model with v = 0.8. The blue lines show the equilibrium trajectory for increasing d_A and the orange lines show the equilibrium trajectory for decreasing d_A . See Table A for the parameters used. Fig D. Hysteresis for increasing resource congestion *q* for the adaptive model with v = 0.7. Equilibrium abundance of pollinator species as a function of the drivers of decline d_A for increasing resource congestion q for the adaptive model with v = 0.7. The blue lines show the equilibrium trajectory for increasing d_A and the orange lines show the equilibrium trajectory for decreasing d_A . See Table A for the parameters used. Fig E. Hysteresis for increasing resource congestion q for the adaptive model with v = 0.6. Equilibrium abundance of pollinator species as a function of the drivers of decline d_A for increasing resource congestion q for the adaptive model with v = 0.6. The blue lines show the equilibrium trajectory for increasing d_A and the orange lines show the equilibrium trajectory for decreasing d_A . See Table A for the parameters used. Section D. Distribution of pollinator persistence. Describes the distribution of pollinator persisitence in different computational experiments. Fig F. The full distribution of relative pollinator abundance for three different rates of change λ accompanying Fig 2A in the paper (no adaptive foraging). θ is the fraction of the point of collapse d_A at which point the relative pollinator persistence is measured. The distributions are bimodal around 0 and 1 which indicates that there is an abrupt collapse of networks at increasing rates of change. See Table A for the parameters used. Fig G. The full distribution of relative pollinator abundance for three different rates of change λ accom**panying** Fig 2A in the paper (with adaptive foraging). θ is the fraction of the point of collapse d_A at which point the relative pollinator persistence is measured. The distributions are mainly bimodal around 0 and 1. However, some networks have a persistence between 0 and 1, indicating partial collapse due to the rate of change. Furthermore, there are a few networks with pollinator persistence significantly above 1, indicating nonlinear effects where sometimes individual networks can profit from higher rates of change. See Table A for the parameters used. Section E. Sensitivity analysis. Contains: Table B. Parameters for the sensitivity analysis on the feasibility of networks. Parameters and their value ranges used for the sensitivity analysis on the feasibility of networks, and plant and pollinator abundances. The fixed parameters can be found in Table A. AF = Adaptive Foraging. Fig H. Sensitivity analysis of the number of plant species alive. Sobol sensitivity analysis of the number of plant species alive depending on five parameters: resource congestion q, nestedness N, connectance D, adaptation strength v, and migration rate μ . The sample size per parameter was 512. The adaptation strength v had the strongest effect on the variance of the outcome of the model. Fig I. Sensitivity analysis of the number of pollinator species alive. Sobol sensitivity analysis of the number of pollinator species alive depending on five parameters: resource congestion q, nestedness N, connectance D, adaptation strength v, and migration rate μ . The sample size per parameter was 512. The adaptation strength v had the strongest effect on the variance of the number of pollinators alive. The migration rate μ only has a marginal effect. Fig J. Sensitivity analysis of the total number of species alive. Sobol sensitivity analysis of the total number of species alive depending on five parameters: resource congestion q, nestedness N, connectance D, adaptation strength v, and migration rate μ . The sample size per parameter was 512. The adaptation strength v had the strongest effect on the variance of the outcome of the model. Fig K.

Sensitivity analysis of the abundance of plant species. Sobol sensitivity analysis of the average plant abundance depending on five parameters: resource congestion q, nestedness N, connectance D, adaptation strength v, and migration rate μ . The sample size per parameter was 512. Fig L. Sensitivity analysis of the abundance of pollinator species. Sobol sensitivity analysis of the average pollinator abundance depending on five parameters: resource congestion q, nestedness N, connectance D, adaptation strength v, and migration rate μ . The sample size per parameter was 512. Table C. Parameters for the sensitivity analysis on the critical driver of decline of collapse $d_A^{collapse}$. Parameters and their value ranges used for the sensitivity analysis on the critical driver of decline of collapse $d_A^{collapse}$. The fixed parameters can be found in Table A. AF = Adaptive Foraging, d_A = driver of decline. Fig M. Sensitivity analysis on the driver of decline d_A . Sobol sensitivity analysis on the value of driver of decline at which all pollinator are extinct $d_A^{collapse}$, depending on six parameters: resource congestion q, nestedness N, connectance D, adaptation strength v, initial abundance per species S^{init} , and migration rate μ . The sample size per parameter was 512. (PDF)

Acknowledgments

The authors acknowledge the fruitful discussions and stimulating environment that occurred during the 'BIRS workshop [22w5067]: Rate-Induced Transitions in Networked Systems.' A special thanks to the several discussions with Guido Vaessen, who provided continuous comments throughout, and to Theresa W. Ong, Kristen M. Jovanelly, Chenyang Su, and Wenying Liao, who provided initial comments on the research. A further thank-you note to Mike Lees, who provided valuable and motivating comments on the research and initial versions of the figures and original manuscript. ST acknowledges the support of the Institute of Advanced Study (University of Amsterdam) where part of the work was carried out.

Author Contributions

Conceptualization: Sjoerd Terpstra, Flávia M. D. Marquitti, Vítor V. Vasconcelos. Formal analysis: Sjoerd Terpstra, Vítor V. Vasconcelos. Funding acquisition: Vítor V. Vasconcelos. Investigation: Sjoerd Terpstra, Vítor V. Vasconcelos. Methodology: Sjoerd Terpstra, Flávia M. D. Marquitti, Vítor V. Vasconcelos. Project administration: Vítor V. Vasconcelos. Software: Sjoerd Terpstra. Supervision: Vítor V. Vasconcelos. Validation: Sjoerd Terpstra, Flávia M. D. Marquitti, Vítor V. Vasconcelos. Visualization: Sjoerd Terpstra, Vítor V. Vasconcelos. Writing – original draft: Sjoerd Terpstra, Vítor V. Vasconcelos. Writing – review & editing: Sjoerd Terpstra, Flávia M. D. Marquitti, Vítor V. Vasconcelos.

References

- Potts S, Biesmeijer J, Kremen C, Neumann P, Schweiger O, Kunin W. Global pollinator declines: Trends, impacts and drivers. Trends in Ecology and Evolution. 2010; 25(6):345–353. <u>https://doi.org/10.1016/j.tree.2010.01.007</u> PMID: 20188434
- Ollerton J, Winfree R, Tarrant S. How many flowering plants are pollinated by animals? Oikos. 2011; 120(3):321–326.
- Vanbergen A, Garratt M, Baude M, Biesmeijer J, Britton N, Brown M, et al. Threats to an ecosystem service: Pressures on pollinators. Frontiers in Ecology and the Environment. 2013; 11(5):251–259. https://doi.org/10.1890/120126
- Eilers E, Kremen C, Greenleaf S, Garber A, Klein A. Contribution of pollinator-mediated crops to nutrients in the human food supply. PLoS ONE. 2011; 6(6). <u>https://doi.org/10.1371/journal.pone.0021363</u> PMID: 21731717
- Klein A, Vaissière B, Cane J, Steffan-Dewenter I, Cunningham S, Kremen C, et al. Importance of pollinators in changing landscapes for world crops. Proceedings of the Royal Society B: Biological Sciences. 2007; 274(1608):303–313. https://doi.org/10.1098/rspb.2006.3721 PMID: 17164193
- Bascompte J, Jordano P, Melián C, Olesen J. The nested assembly of plant-animal mutualistic networks. Proceedings of the National Academy of Sciences of the United States of America. 2003; 100 (16):9383–9387. https://doi.org/10.1073/pnas.1633576100 PMID: 12881488
- Aizen M, Morales C, Morales J. Invasive mutualists erode native pollination webs. PLoS Biology. 2008; 6(2):0396–0403. https://doi.org/10.1371/journal.pbio.0060031 PMID: 18271628
- Medan D, Perazzo R, Devoto M, Burgos E, Zimmermann M, Ceva H, et al. Analysis and assembling of network structure in mutualistic systems. Journal of Theoretical Biology. 2007; 246:510–521. <u>https://doi.org/10.1016/j.jtbi.2006.12.033</u> PMID: 17316696
- Atmar W, Patterson B. The measure of order and disorder in the distribution of species in fragmented habitat. Oecologia. 1993; 96(3):373–382. https://doi.org/10.1007/BF00317508 PMID: 28313653
- Burkle L, Marlin J, Knight T. Plant-Pollinator Interactions over 120 Years: Loss of Species, Co-Occurrence, and Function. Science. 2013; 339:1611–1615. https://doi.org/10.1126/science.1232728 PMID: 23449999
- Sánchez-Bayo F, Wyckhuys K. Worldwide decline of the entomofauna: A review of its drivers. Biological Conservation. 2019; 232:8–27. https://doi.org/10.1016/j.biocon.2019.01.020
- Potts S, Ngo H, Biesmeijer J, Breeze T, Dicks L, Garibaldi L, et al. The assessment report of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services on pollinators, pollination and food production. Secretariat of the Intergovernmental Science-Policy Platform on Biodiversity; 2016.
- Dicks L, Breeze T, Ngo H, Senapathi D, An J, Aizen M, et al. A global-scale expert assessment of drivers and risks associated with pollinator decline. Nature Ecology and Evolution. 2021; 5:1453–1461. https://doi.org/10.1038/s41559-021-01554-5 PMID: 34400826
- Vasiliev D, Greenwood S. The role of climate change in pollinator decline across the Northern Hemisphere is underestimated. Science of the Total Environment. 2021; 775. https://doi.org/10.1016/j. scitotenv.2021.145788 PMID: 33618305
- Lamberson P. Tipping Points. Quarterly Journal of Political Science. 2012; 7:175–208. https://doi.org/ 10.1561/100.00011061
- Lenton T, Held H, Kriegler E, Hall J, Lucht W, Rahmstorf S, et al. Tipping elements in the Earth System. Proceedings of the National Academy of Sciences of the United States of America. 2008; 105:1786– 1793. https://doi.org/10.1073/pnas.0911106106
- van Nes E, Arani B, Staal A, van der Bolt B, Flores B, Bathiany S, et al. What Do You Mean, 'Tipping Point'? Trends in Ecology and Evolution. 2016; 31:902–904. <u>https://doi.org/10.1016/j.tree.2016.09.011</u> PMID: 27793466
- 18. Scheffer M. Critical Transitions in Nature and Society. vol. 452. Princeton Unversity Press; 2009.
- Feudel U, Pisarchik A, Showalter K. Multistability and tipping: From mathematics and physics to climate and brain—Minireview and preface to the focus issue. Chaos. 2018; 28. <u>https://doi.org/10.1063/1.5027718</u>
- Krönke J, Wunderling N, Winkelmann R, Staal A, Stumpf B, Tuinenburg OA, et al. Dynamics of tipping cascades on complex networks. Physical Review E. 2020; 101:1–9. PMID: 32422827
- **21.** Rockström J, Steffen W, Noone J, Persson Å, Chapin F III, Lambin E, et al. Planetary boundaries: exploring the safe operating space for humanity. Ecology and society. 2009; 14(2).
- Kemp L, Xu C, Depledge J, Ebi K, Gibbins G, Kohler T, et al. Climate Endgame: Exploring catastrophic climate change scenarios. Proceedings of the National Academy of Sciences. 2022; 119(34): e2108146119. https://doi.org/10.1073/pnas.2108146119 PMID: 35914185

- Latty T, Dakos V. The risk of threshold responses, tipping points, and cascading failures in pollination systems. Biodiversity and Conservation. 2019; 28:3389–3406. https://doi.org/10.1007/s10531-019-01844-2
- Bryden J, Gill R, Mitton R, Raine N, Jansen V. Chronic sublethal stress causes bee colony failure. Ecology Letters. 2013; 16:1463–1469. https://doi.org/10.1111/ele.12188 PMID: 24112478
- Dakos V, Bascompte J. Critical slowing down as early warning for the onset of collapse in mutualistic communities. Proceedings of the National Academy of Sciences of the United States of America. 2014; 111:17546–17551. https://doi.org/10.1073/pnas.1406326111 PMID: 25422412
- Lever J, van Nes E, Scheffer M, Bascompte J. The sudden collapse of pollinator communities. Ecology Letters. 2014; 17:350–359. https://doi.org/10.1111/ele.12236 PMID: 24386999
- Meng Y, Grebogi C. Control of tipping points in stochastic mutualistic complex networks. Chaos. 2021; 31. https://doi.org/10.1063/5.0036051 PMID: 33653048
- 28. Jiang J, Hastings A, Lai Y. Harnessing tipping points in complex ecological networks. Journal of the Royal Society Interface. 2019; 16. https://doi.org/10.1098/rsif.2019.0345 PMID: 31506040
- Huang H, Tu C, D'Odorico P. Ecosystem complexity enhances the resilience of plant-pollinator systems. One Earth. 2021; 4:1286–1296. https://doi.org/10.1016/j.oneear.2021.08.008
- Siteur K, Eppinga M, Doelman A, Siero E, Rietkerk M. Ecosystems off track: rate-induced critical transitions in ecological models. Oikos. 2016; 125:1689–1699. https://doi.org/10.1111/oik.03112
- Ashwin P, Wieczorek S, Vitolo R, Cox P. Tipping points in open systems: bifurcation, noise-induced and rate-dependent examples in the climate system. Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences. 2012; 370:1166–1184. https://doi.org/10.1098/rsta. 2011.0306 PMID: 22291228
- Petanidou T, Kallimanis A, Tzanopoulos J, Sgardelis S, Pantis J. Long-term observation of a pollination network: Fluctuation in species and interactions, relative invariance of network structure and implications for estimates of specialization. Ecology Letters. 2008; 11(6):564–575. <u>https://doi.org/10.1111/j.</u> 1461-0248.2008.01170.x PMID: 18363716
- Ponisio L, Gaiarsa M, Kremen C. Opportunistic attachment assembles plant–pollinator networks. Ecology Letters. 2017; 20(10):1261–1272. https://doi.org/10.1111/ele.12821 PMID: 28921857
- 34. Hagen M, Kissling W, Rasmussen C, Aguiar MD, Brown L, Carstensen D, et al. Biodiversity, Species Interactions and Ecological Networks in a Fragmented World. vol. 46. Academic Press Inc.; 2012.
- Noreika N, Bartomeus I, Winsa M, Bommarco R, Öckinger E. Pollinator foraging flexibility mediates rapid plant-pollinator network restoration in semi-natural grasslands. Scientific Reports. 2019; 9(1):1– 11. https://doi.org/10.1038/s41598-019-51912-4 PMID: 31664170
- Valdovinos F, Moisset de Espanés P, Flores J, Ramos-Jiliberto R. Adaptive foraging allows the maintenance of biodiversity of pollination networks. Oikos. 2013; 122(6):907–917. <u>https://doi.org/10.1111/j.</u> 1600-0706.2012.20830.x
- Holling C. The Components of Predation as Revealed by a Study of Small-Mammal Predation of the European Pine Sawfly. The Canadian Entomologist. 1959; 91:293–320. https://doi.org/10.4039/ Ent91293-5
- Olesen J, Bascompte J, Dupont Y, Elberling H, Rasmussen C, Jordano P. Missing and forbidden links in mutualistic networks. Proceedings of the Royal Society B: Biological Sciences. 2011; 278 (1706):725–732. https://doi.org/10.1098/rspb.2010.1371 PMID: 20843845
- Tilman A, Dixit S AKand Levin. Localized prosocial preferences, public goods, and common-pool resources. Proceedings of the National Academy of Sciences. 2019; 116(12):5305–5310. <u>https://doi.org/10.1073/pnas.1802872115</u> PMID: 30297394
- 40. Virtanen P, Gommers R, Oliphant T, Haberland M, Reddy T, Cournapeau D, et al. SciPy 1.0: fundamental algorithms for scientific computing in Python. Nature Methods. 2020; 17:261–272. <u>https://doi.org/10. 1038/s41592-019-0686-2 PMID: 32015543</u>
- **41.** Holling C. Resilience and stability of ecological systems. The Future of Nature: Documents of Global Change. 1973; 4:245–256.
- Aizen M, Sabatino M, Tylianakis J. Specialization and rarity predict nonrandom loss of interactions from mutualist networks. Science. 2012; 335(6075):1486–1489. https://doi.org/10.1126/science.1215320 PMID: 22442482
- Lomáscolo S, Giannini N, Chacoff N, Castro-Urgal R, Vázquez D. Inferring coevolution in a plant–pollinator network. Oikos. 2019; 128:775–789. https://doi.org/10.1111/oik.05960
- Opedal Ø. The evolvability of animal-pollinated flowers: towards predicting adaptation to novel pollinator communities. New Phytologist. 2019; 221(2):1128–1135. <u>https://doi.org/10.1111/nph.15403</u> PMID: 30145801

- Ponisio L, Valdovinos F, Allhoff K, Gaiarsa M, Barner A, Guimarães P, et al. A network perspective for community assembly. Frontiers in Ecology and Evolution. 2019; 7(APR).
- P L, M I, L M, R M, R MC. Rate of environmental change across scales in ecology. Biological Reviews. 2020; 95:1798–1811. https://doi.org/10.1111/brv.12639
- Ramos S, Schiestl F. Rapid plant evolution driven by the interaction of pollination and herbivory. Science. 2019; 364:193–196. https://doi.org/10.1126/science.aav6962 PMID: 30975889
- Farrel BD, Mitter C. Adaptive Radiation in Insects and Plants: Time and Opportunity. American Zoologist. 1994; 34:57–69. https://doi.org/10.1093/icb/34.1.57
- Gervasi D, Schiestl F. Real-time divergent evolution in plants driven by pollinators. Nature Communications. 2017; 8:14691. https://doi.org/10.1038/ncomms14691 PMID: 28291771
- Kaiser-Bunbury C, Muff S, Memmott J, Müller C, Caflisch A. The robustness of pollination networks to the loss of species and interactions: A quantitative approach incorporating pollinator behaviour. Ecology Letters. 2010; 13:442–452. https://doi.org/10.1111/j.1461-0248.2009.01437.x PMID: 20100244
- Sheykhali S, Fernández-Gracia J, Traveset A, Ziegler M, Voolstra C, Duarte C, et al. Robustness to extinction and plasticity derived from mutualistic bipartite ecological networks. Scientific Reports. 2020; 10(1):1–12. https://doi.org/10.1038/s41598-020-66131-5
- Puy A, Beneventano P, Levin SA, Piano S, Portaluri T, Saltelli A. Models with higher effective dimensions tend to produce more uncertain estimates. Science Advances. 2022; 8:9450. https://doi.org/10. 1126/sciadv.abn9450 PMID: 36260678
- Convention on Biological Diversity C. Recommendation Adopted By the Working Group on the Post-2020 Global Diversity Framework, Outcomes of Part II of the Third Meeting of the Working Group; 2022. Available from: https://www.cbd.int/doc/recommendations/wg2020-03/wg2020-03-rec-01-en.pdf.
- Buschke F, Brownlie S. Reduced ecological resilience jeopardizes zero loss of biodiversity using the mitigation hierarchy. Nature Ecology & Evolution. 2020; 4(6):815–819. https://doi.org/10.1038/s41559-020-1177-7 PMID: 32251380
- 55. Ramos-Jiliberto R, de Espanés P, Vázquez D. Pollinator declines and the stability of plant–pollinator networks. Ecosphere. 2020; 11. https://doi.org/10.1002/ecs2.3069
- Bascompte J, Scheffer M. The resilience of plant–pollinator networks. Annual Review of Entomology. 2023; 68:363–380. https://doi.org/10.1146/annurev-ento-120120-102424 PMID: 36206771